Author's response to reviews

Title: Acute Bilateral Simultaneous Angle Closure Glaucoma after Topiramate Administration

Authors:

Kakarla V Chalam (kchalam@jax.ufl.edu)
Tina Tillis (tina.tillis@jax.ufl.edu)
Farhana Syed (farhana.syed@jax.ufl.edu)
Agarwal Swati (swati.agarwal@jax.ufl.edu)
Vikram S Brar (vikram.brar@jax.ufl.edu)

Version: 2 Date: 25 July 2007

Author's response to reviews: see over
July 25, 2007

Editor-in-Chief
Michael Kidd, M.D.

Dear Dr. Norton,

We are submitting a case report entitled “Acute Bilateral Simultaneous Angle Closure Glaucoma after Topiramate Administration” for consideration of publication in BMC Ophthalmology as a research article.

Topiramate, an oral sulpha-derivative medication is used commonly to treat migraine and epilepsy. This report highlights the precipitation of bilateral acute angle closure glaucoma with severe visual loss in a patient immediately after initiation of treatment for migraine.

The manuscript has been seen and approved by all authors and that it is not under consideration for publication elsewhere in a similar form, in any language, except in abstract form. Patient consent was obtained for submitted figures containing clinical photographs.

We do not have any financial interests in this work.

Looking forward to your positive response.

Sincerely,

K. V. Chalam, MD, PhD, MBA, FACS
Professor and Chairman
Department of Ophthalmology, Health Science Center
University of Florida, Jacksonville Campus
Jacksonville, Florida.